Author’s response to reviews

Title: Exercise to preserve beta cell function in recent-onset type 1 diabetes mellitus (EXTOD)-a study protocol for a pilot randomised controlled trial

Authors:

Nadia Lascar (n.lascar@bham.ac.uk)
Amy Kennedy (a.kennedy@bham.ac.uk)
Nikki Jackson (cdaj@bris.ac.uk)
Amanda Daley (a.daley@bham.ac.uk)
George Dowswell (g.dowswell@bham.ac.uk)
Dylan Thompson (d.thompson@bath.ac.uk)
Keith Stokes (k.stokes@bath.ac.uk)
Sheila Greenfield (s.m.greenfield@bham.ac.uk)
Roger Holder (r.l.holder@bham.ac.uk)
Rob Andrews (rob.andrews@bris.ac.uk)
Parth Narendran (p.narendran@bham.ac.uk)

Version: 2 Date: 15 May 2013

Author’s response to reviews:

Thank you for your email of 14th May regarding our manuscript, and the very positive feedback provided by your referee.

We have addressed their points in the attached cover letter.

Kind regards

Parth Narendran